Author's response to reviews

Title: The effects of oral iron supplementation on cognition in older children and adults: a systematic review and meta-analysis

Authors:

Martin Falkingham (martinfalkingham@googlemail.com)
Asmaa Abdelhamid (Asmaa.abdelhamid@uea.ac.uk)
Peter Curtis (p.curtis@uea.ac.uk)
Susan Fairweather-Tait (s.fairweather-tait@uea.ac.uk)
Louise Dye (l.dye@leeds.ac.uk)
Lee Hooper (l.hooper@uea.ac.uk)

Version: 6 Date: 20 January 2010

Author's response to reviews:

Hi, one very minor change. A systematic review mentioned in the discussion as under review has now been published, so the reference has now been added (Hoyland et al, 2009).

Thank you for your help with this manuscript! all the best, Lee (for the authors)